

Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2021GL095574

Key Points:

- We assess the physical grounding of Atlantic-European winter weather regimes (WR) using dynamical systems (DS) theory
- Most WR display distinct and predictable flow characteristics at their maximum stage, and unpredictable flow at onset and decay
- The agreement between the statistical classification and the DS analysis provides strong evidence for most WR being physically meaningful

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

A. Hochman, assaf.hochman@kit.edu; assaf.hochman@mail.huji.ac.il

Citation:

Hochman, A., Messori, G., Quinting, J. F., Pinto, J. G., & Grams, C. M. (2021). Do Atlantic-European weather regimes physically exist? *Geophysical Research Letters*, 48, e2021GL095574. https://doi. org/10.1029/2021GL095574

Received 4 AUG 2021 Accepted 17 SEP 2021

Author Contributions:

Conceptualization: Assaf Hochman, Gabriele Messori, Julian F. Quinting, Christian M. Grams Investigation: Assaf Hochman, Gabriele Messori, Julian F. Quinting, Christian M. Grams Methodology: Assaf Hochman, Gabriele Messori, Christian M. Grams Supervision: Joaquim G. Pinto Writing – original draft: Assaf Hochman

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Do Atlantic-European Weather Regimes Physically Exist?

Assaf Hochman^{1,2}, Gabriele Messori^{3,4}, Julian F. Quinting¹, Joaquim G. Pinto¹, and Christian M. Grams¹

¹Department of Tropospheric Research (IMK-TRO), Institute of Meteorology and Climate Research, Karlsruhe Institute of Technology (KIT), Karlsruhe, Germany, ²Fredy and Nadine Hermann Institute of Earth Sciences, The Hebrew University of Jerusalem (HUJI), Jerusalem, Israel, ³Department of Earth Sciences and Centre of Natural Hazards and Disaster Science (CNDS), Uppsala University, Uppsala, Sweden, ⁴Department of Meteorology and Bolin Centre for Climate Research, Stockholm University, Stockholm, Sweden

Abstract The subseasonal variability of the extratropical large-scale atmospheric flow is characterized by recurrent or quasi-stationary circulation anomalies, termed weather regimes. Despite the usefulness of these regimes in numerous meteorological and socioeconomic applications, there is an ongoing debate as to whether they represent physical modes of the atmosphere, or are merely useful statistical categorizations. Here, we answer this question for wintertime Atlantic-European regimes. We argue that dynamical systems theory applied to a refined regime definition provides strong evidence in support of most weather regimes being physically meaningful. This finding underpins the broad relevance of weather regimes, for understanding the response of the atmosphere to external forcing, supporting subseasonal weather forecasting, and down scaling of climate projections.

Plain Language Summary The weather situation over the North Atlantic and Europe region is dominated by steady large-scale atmospheric circulation patterns that occur repeatedly, often termed weather regimes. Despite their usefulness in weather forecasting, it is questioned whether these regimes have a clear physical grounding, or are merely useful statistical classifications. By showing that the atmospheric circulation indeed settles down in a comparably steady state during certain regimes, we provide strong evidence in support of their physical foundation. This finding underpins the broad relevance of weather regimes in supporting weather forecasting several weeks ahead and, in addition, understanding the impacts of climate change on our atmosphere.

1. Introduction

The subseasonal atmospheric variability in the extratropics is a key research topic within atmospheric science and of broad socioeconomic relevance (White et al., 2017). A far-reaching paradigm is that a small number of states, termed "weather regimes," can describe this variability. These are defined as recurrent or quasi-stationary states of the large-scale circulation (Hannachi et al., 2017). Weather regimes have implications for extended-range weather forecasting and in the understanding of climate variability (Merryfield et al., 2020).

The concept of weather regimes was first introduced by weather forecasters in the late 1940s (Levick, 1949). A corresponding theory was developed by Charney and DeVore (1979), who hypothesized that the largescale circulation transitions between multiple equilibria states, based on a spectral model. The multiple equilibria viewpoint has been later authenticated for the barotropic case (Legras & Ghil, 1985), and in two-layer models (Yoden, 1983), yet was criticized using models with higher spectral resolution (Cehelsky & Tung, 1987). Faranda et al. (2016) argued for linking blocked regimes to unstable fixed points rather than stable equilibria. Nowadays, there is little doubt that weather regimes emerge from different statistical classifications applied to long archives (Hannachi et al., 2017). However, it is a source of debate whether these regimes represent metastable states of the atmosphere (Majda et al., 2006), or are merely useful statistical categorizations lacking physical grounding (Fereday, 2017).

A variety of procedures has been used to classify weather regimes (Hannachi et al., 2017). They typically seek to define a low-dimensional phase-space reflecting the key aspects of the atmosphere's variability. A common choice is to compute the Empirical Orthogonal Functions (EOF) of the data. Clustering algorithms



Writing – review & editing: Assaf Hochman, Gabriele Messori, Julian F. Quinting, Joaquim G. Pinto, Christian M. Grams define weather regimes as distinct states of the atmosphere that have a high probability of occurrence and are separated by transitional lower probability states. Several clustering techniques exist—the most common being the *k*-means approach (Michelangeli et al., 1995)—aiming to provide partitions of the phase-space into regions, to which datapoints are assigned. A difficulty with cluster analyses is determining the a priori ideal number of clusters (Falkena et al., 2020).

Different numbers of regimes based on geopotential height were proposed to characterize Atlantic-European weather variability. For example, the North Atlantic Oscillation (NAO) provides a two-state classification (Wallace & Gutzler, 1981). Other classifications use 4 (Vautard, 1990) or 6 classes (Falkena et al., 2020). Classifications based on the zonal wind, yield 3-5 jet regimes (Dorrington & Strommen, 2020; Woollings et al., 2010), with four jet regimes corresponding to the four classical weather regimes (Madonna et al., 2017). However, neither are these definitions applicable year-round nor may the number of regimes sufficiently describe the intraseasonal weather variability in the region (Grams et al., 2017). Here, we use the year-round weather regime classification of Grams et al. (2017). This is based on combining EOF analysis and *k*-means clustering and identifies seven regimes.

We approach the discussion of the weather regimes' physical grounding by leveraging recent advances in dynamical systems theory. These allow to characterize instantaneous atmospheric states in terms of the local dimension (*d*)—which informs on how the atmosphere evolves to and from a given state—and persistence in phase space (θ^{-1} ; Faranda et al., 2017). These metrics are related to the *intrinsic predictability* of the atmosphere and therefore to the *stability* (here in the context of atmospheric variability) of the flow: a highly persistent (low θ), low-dimensional (low *d*) state will be more stable than a low-persistence (high θ), high-dimensional (high *d*) one (Messori et al., 2017). This approach has been applied to various atmospheric fields (e.g., Faranda et al., 2019a; Messori et al., 2021). Indeed, it has been shown that *d* and θ can offer a dynamical characterization of synoptic systems over several geographical regions (Faranda, Alvarez-Castro, et al., 2017; Hochman et al., 2019; Hochman, Alpert, et al., 2020). These studies have focused on individual time-slices of the atmosphere. Yet, recent studies have demonstrated the potential of using the temporal evolution of the metrics to study the predictability of extreme weather events (Hochman, Scher, et al., 2020; Hochman et al., 2021).

We leverage the dynamical systems approach to answer the question: Do weather regimes have a strong physical grounding in the dynamics of the atmospheric circulation, or are they simply convenient statistical categorizations? The governing hypothesis is that if weather regimes are dynamically meaningful states, they should demonstrate unstable flow characteristics (high d and θ) before their onset and after their decay, and relatively stable flow (low d and θ) during their mature stage.

2. Materials and Methods

2.1. Data

The Atlantic-European weather regime life cycle classification and the dynamical systems analysis are based on 6-hourly, 0.7° horizontal resolution European Centre for Medium-Range Weather Forecast (EC-MWF) ERA-Interim reanalysis data for 1979–2019 (Dee et al., 2011). The analysis is performed on 500-hPa geopotential height (Z500) for the Atlantic-European region (80°W–40°E, 30–90°N; Figure 1). This variable is routinely used to diagnose the large-scale atmospheric flow characteristics (Ghil & Robertson, 2002).

2.2. Weather Regime Life Cycle

We use the year-round Atlantic-European weather regime definition of Grams et al. (2017). First, we perform a *k*-means clustering in the phase space spanned by the leading seven EOFs (~76% explained variance) of 6-hourly (1979–2015) 10-day low-pass filtered, normalized Z500 anomalies (Z500') over the Atlantic-European region. Data is interpolated to a 1.0° grid spacing. Next, we compute the normalized projection of each 6-hourly Z500' onto the cluster mean, which we term the Weather Regimes Index (*IWR*; Michel & Rivière, 2011). The *IWR* time series is extended until February 2019. We then define weather regime life cycles. The key characteristic of an active life cycle for a regime *w* is that at the time of onset, the respective *IWR*_w exceeds the value of 1.0 for the first time. Subsequently, *IWR*_w must remain above 1.0 for \geq 5 days. At





Figure 1. Atlantic-European weather regimes in winter. (a–g) 500-hPa geopotential height mean composites (contours, every 80 gpm) and their anomalies (shading, every 20 gpm) with respect to the seasonal climatology (h) during active regime life cycles (the onset to decay period) in December, January, February of 1979–2019. The weather regimes considered are: (a) Atlantic Trough (AT), (b) Zonal Regime (ZO), (c) Scandinavian Trough (ScTr), (d) Atlantic Ridge (AR), (e) European Blocking (EuBL), (f) Scandinavian Blocking (ScBL), (g) Greenland Blocking (GL). Numbers in the subfigure titles indicate the frequencies of winter days attributed to the respective regime (a–g). The frequency of "no regime" days is 24.1%.

the maximum stage, IWR_w reaches its highest value and it then falls below 1.0 in the decay stage. Still, the projection IWR_i for any other regime *i* can be computed at each time and reflects the concurrent contribution of different regimes to the circulation pattern.

Dates with all *IWR* values below 1.0 or none of the *IWR*_w fulfilling the life cycle criteria have "no regime" (24.1% of winter days). Negative Z500′ dominates in three out of the seven weather regimes ("Atlantic Trough"—AT; "Zonal Regime"—ZO; "Scandinavian Trough"—ScTr), which are termed "cyclonic" regimes (Figures 1a–1c). The remaining regimes are dominated by positive Z500′ and are termed "blocked regimes" ("Atlantic Ridge"—AR; "European Blocking"—EuBL; "Scandinavian Blocking"—ScBL; "Greenland Blocking"—GL; Figures 1d–1g). For more details on the regime characteristics, the reader is referred to Grams et al. (2017). This study focuses on weather regime life cycles being active in the winter months (DJF); summer months (JJA) are briefly discussed. The Kolmogorov-Smirnov test is used for assessing the differences between regime distributions at the 5% significance level.

2.3. Dynamical Systems Analysis

To depict the dynamical evolution of Atlantic-European weather regimes, we rely on an approach combining extreme value theory with Poincaré recurrences (Lucarini et al., 2012, 2016). This approach allows the computation of instantaneous properties of chaotic dynamical systems, and hence it is ideally suited to study the dynamics of the atmosphere. We interpret a temporal succession of two-dimensional Z500 maps as discrete samples from a long trajectory in a reduced atmospheric phase space. For each map, we compute two instantaneous dynamical properties, which are termed local dimension (*d*) and persistence (θ^{-1}).

The local dimension (d) is a proxy for the active number of degrees of freedom that a system can explore locally. It therefore reflects the way the system approaches and departs from a given atmospheric state. The computation of d stems from the fact that the cumulative probability distribution of suitably defined recurrences of the system converges to the exponential member of the Generalized Pareto Distribution (Freitas et al., 2010).

The persistence (θ^{-1}) of a state intuitively relates to the persistence time of the system near said state. θ^{-1} tends to be more sensitive to small changes in the state of the system than other persistence metrics, such as the conventional definition of weather regime persistence. In the latter

case, we are partitioning the atmospheric variability into a small number of states, while in the former we define "similar" states as a small number of close recurrences. Notwithstanding these differences, the two definitions show a close relationship (Hochman et al., 2019). To compute θ^{-1} , we estimate the extremal index using the Süveges (2007) estimator. For details of the computation of the metrics, the reader is referred to Faranda, Messori and Vannistem (2019) and Messori et al. (2017).

Studies have shown that the dynamical systems metrics have a strong seasonal cycle (Faranda et al., 2017; Hochman, Scher, et al., 2020). Since we are comparing weather regime life cycles during different parts of the winter season, we deseasonalize the metrics prior to our analysis. The seasonal cycle is computed by averaging the metrics for a given time step over all years, repeating this for all time steps within the year and finally smoothing the series with a 30-day moving average. A bootstrap test is used to estimate the 95% confidence intervals for the mean temporal evolutions of the metrics. The same test as in Section 2.2 is used for comparing the metrics distributions.





Figure 2. Average temporal evolution of the objective Weather Regimes Index (*IWR*) centered on the weather regime maximum stage (lag on x-axis = 0 days) during the respective active life cycle. Panels (a)–(g) display the mean *IWR* of all regimes during active regime life cycles (*IWR*_w, see subtitles) of the (a) Atlantic Trough (AT), (b) Zonal Regime (ZO), (c) Scandinavian Trough (ScTr), (d) Atlantic Ridge (AR), (e) European Blocking (EuBL), (f) Scandinavian Blocking (ScBL), (g) Greenland Blocking (GL) regimes. (h) The mean temporal evolution of the respective *IWR*_w if the weather regimes are active. Very bold sections in the curves for *IWR*_w of the active regime indicate significant difference of the Cumulative Distribution Functions (CDF) of *IWR*_w at the 5% level from the CDF of all other regimes (*IWR*_i; see text). Bold sections in the curves for *IWR*_i indicate significant differences of the CDF of *IWR*_i to the CDF of the active regime's *IWR*_w.

3. Results

3.1. Atlantic-European Weather Regimes Life Cycles

First, we analyze the temporal evolution of the mean IWR_w during the active life cycle of regime w, and IWR_i , for all other regimes *i*. The composites are centered on the maximum stage of regime w (Figures 2a–2g). At the time of the maximum (0 days) the mean IWR_w (very bold curves; Figures 2a–2g) is distinct from all other mean IWR_i (bold curves). The distributions of IWR_w over a time window of about ±5 days around

the maximum stage are significantly different from the distributions of IWR_i over the same period (very bold sections of IWR_w ; Figures 2a–2g). Thus, the atmospheric state in a time window around the maximum stage of a regime is distinct from all other regime states, which suggests a coherent life-cycle evolution, and a low local dimension in terms of dynamical systems theory.

The period of mean $IWR_w > 1.0$ is related to the duration of a regime life cycle, and fulfilled on average for ≥ 9 days for all regimes (Figure 2h), while the persistence criterion in the life cycle definition is ≥ 5 days. The mean regime duration in winter, i.e., the mean period between the identified regime onset and decay for each regime life cycle ranges from 10.8 days for EuBL to 16.9 days for GL (not shown). Contrary to what might be expected, the blocked EuBL (10.8 days) and ScBL (11.0 days) regimes display lower mean duration than the cyclonic regimes (13.3–14.6 days).

To further determine whether the IWR_{w} values are different from IWR_{i} at the maximum stage, we evaluate their Cumulative Distribution Functions (CDFs; Figures 3a-3g). The CDFs for some regimes tend to be closer to the CDF of the active regime (rightmost CDF; Figures 3a-3g) than others (left CDFs; Figures 3a-3g). During an active ZO regime, the CDFs of the AT and ScTr are closer to the CDF of ZO than the blocked regimes (Figure 3b). This similarity between ZO, AT, and ScTr is also reflected in the mean Z500 patterns, which are characterized by negative anomalies over the Eastern North Atlantic (Figures 1a-1c). During a ScTr regime, the CDFs of the cyclonic regimes, but also of the AR regime are closest to the ScTr (Figure 3c). Vice-versa during an active AR regime, the closest CDFs are those of ScTr, GL, and EuBL (Figure 3d). As a final example, the CDFs of the EuBL and ScBL regimes are closest to each other no matter which regime is active (Figures 3e and 3f). The CDF for GL shows higher IWR, values than all other regimes, and GL has a longer life cycle. On the other hand, the CDF for ScBL exhibits the overall lowest IWR, values during its active life cycle (cf. Figures 2h and 3h). It is nonetheless important to emphasize that the active regime CDF is always significantly different from all the others (the CDFs to the left) and this difference persists for about ± 5 days around the maximum stage (Figure 2). Thus, the atmospheric state at maximum stage of a regime is distinct from all other regime states, again suggesting a low local dimension in terms of dynamical systems theory.

3.2. Atlantic-European Weather Regimes From a Dynamical Systems Perspective

Next, we characterize the weather regime life cycles directly in terms of the two dynamical systems metrics: local dimension (d) and inverse persistence (θ ; Figures 4 and 5). Both metrics are presented as deviations from climatology, thus values below zero indicate a more stable and persistent state compared to climatology. We first discuss the temporal evolution of d and θ centered on the maximum stage (0 day; Figure 4). The metrics are in phase with one another for most regimes. The cyclonic weather regimes and GL display values of d and θ that are significantly below climatology at maximum stage, thus, indicating a low local dimension and enhanced persistence during an active weather regime. The other blocked regimes, except for θ in EuBL and d in ScBL, also display values below climatology, though not significantly so at maximum stage. In addition, d and θ are typically below climatology for several days around the maximum stage. This implies that most weather regimes can be regarded as stable and persistent states from a dynamical systems perspective. In particular, in the cyclonic (Figures 4a-4c) and the blocked AR and GL regimes (Figures 4d and 4g), d and θ start to steadily decrease about 5 days prior to the maximum stage and steadily increase in the 5 days thereafter. This pattern is also found for d in the EuBL regime (Figure 4e). Beyond ± 5 days, d and θ exhibit a larger spread around their climatological values. This behavior of *d* and θ mirrors the onset and decay stages of the active regime life cycle (Figure 2), and is also evident when centering the plots on the onset or decay of the regimes (Figures S1 and S2 in Supporting Information S1). The evolution of d and θ is remarkably consistent with the earlier finding that IWR_w is above 1.0 and significantly different from the other IWR_i in a time window of about ± 5 days around the maximum stage (Figure 2).

Some differences between the regimes deserve further attention. GL is the regime with the most pronounced change in amplitude of both *d* and θ (Figure 4g), indicating the transition from an atmospheric state of low persistence and predictability to a persistent and predictable state. Further, θ is below climatology from 5 days prior to maximum stage until beyond 10 days thereafter, consistent with GL being the regime with the longest mean duration. The cyclonic regimes show a more symmetric evolution of *d* and θ compared to the others, and *d* and θ remain below climatology for around ±3 days (Figures 4a–4c). AR exhibits a roughly





Figure 3. Cumulative Distribution Functions (CDFs) of the objective Weather Regimes Index (*IWR*) at the maximum stage (lag = 0 days in Figure 2) of the Atlantic-European weather regimes. The active weather regimes considered are: (a) Atlantic Trough (AT), (b) Zonal Regime (ZO), (c) Scandinavian Trough (ScTr), (d) Atlantic Ridge (AR), (e) European Blocking (EuBL), (f) Scandinavian Blocking (ScBL), (g) Greenland Blocking (GL). (h) CDFs for all weather regimes at the times when they are active. Significant CDF differences between the active regime and all other regimes are marked with p < 0.05 (a–g). The differences between the weather regimes in (h) significant at the 5% level are shown in Table S1 in Supporting Information S1.

symmetric decrease in *d* and θ toward the maximum stage and a subsequent increase, but the values fall below climatology only for a short period (Figure 4d). EuBL shows similar behavior for *d* (Figure 4e). Of all regimes, θ values remain highest during EuBL, which is consistent with the overall shortest mean duration. ScBL displays the least systematic changes and largest spread in *d* and θ ; though a decrease toward maximum stage is still evident (Figure 4f). This may partly be due to the rarity of ScBL during winter (Figure 1f), giving a small sample size. The fact that *d* falls below climatology only for a short period around maximum stage for EuBL, AR, and remains above climatology for ScBL, hints that the statistical characterization of these regimes has a weaker link to the underlying dynamics of the atmospheric flow.

Finally, we analyze the CDFs of *d* and θ at maximum stage of the regimes (Figure 5). For AT, ZO, ScTr, GL, and to some degree AR, the CDFs for both *d* and θ shift toward values well below climatology, reflecting







Figure 4. Average temporal evolution of the deseasonalized dynamical systems metrics (d and θ) centered on the weather regime *maximum* stage (time = 0 days). The dynamical systems metrics are computed on 500-hPa geopotential height (Z500). A 95% bootstrap confidence interval is shown in shading. The weather regimes considered are: (a) Atlantic Trough (AT), (b) Zonal Regime (ZO), (c) Scandinavian Trough (ScTr), (d) Atlantic Ridge (AR), (e) European Blocking (EuBL), (f) Scandinavian Blocking (ScBL), and (g) Greenland Blocking (GL).

the low local dimension and enhanced persistence at maximum stage. In contrast, the local dimension of ScBL and persistence of EuBL at maximum stage are close to climatology. The CDFs of d and θ display a relatively large spread around climatology at onset and decay stages (not shown).

Though not being the focus of this study, our findings remain qualitatively valid also in summer (Figures S3–S5 in Supporting Information S1). The metrics are in phase with one another for most regimes. The AT, ScBL, and GL display values of d and θ that are significantly below climatology at maximum stage (Figures S4a, S4f, S4g, and S5 in Supporting Information S1). The other regimes, display a minimum value at the maximum stage of a regime, though not significantly different from climatology. The most pronounced difference from the winter season is found for ScBL (cf. Figures S4f in Supporting Information S1 with Figure 4f). ScBL is more frequent in summer (16.0% of all days) compared to winter (6.5% of all days), and in the former season displays d and θ values that are significantly below climatology at maximum stage (Figures S4f and S5 in Supporting Information S1). This strengthens our interpretation of the winter composite as being affected by small sample size.

In summary, our analysis provides strong evidence that most Atlantic-European weather regimes have a dynamical grounding (Figures 4 and 5), which is largely consistent with the statistical-based weather regime life cycle definition (Figures 2 and 3). In winter, this is most pronounced for the cyclonic regimes, as well as for the GL blocked regime.

4. Discussion

We combine a refined weather regime classification with recent advances in dynamical systems theory to investigate the life cycle and physical grounding of Atlantic-European weather regimes. The agreement between the statistical classification and the dynamical systems analysis provides strong evidence for most Atlantic-European weather regimes being physically meaningful. Interestingly, cyclonic regimes and GL, which corresponds to the negative phase of the NAO, show enhanced flow stability and persistence whereas for the blocked regimes of EuBL, ScBL, and AR local dimension reduces toward maximum stage but remains around climatology. Indeed, GL is the single regime with the lowest local dimension and highest persistence, while the EuBL and ScBL regimes emerge as being amongst the lowest-stability regimes. This may be related to the difficulty in predicting these regimes in extended-range

forecasts (Rodwell et al., 2013). The counter-intuitive finding of unstable blocking regimes agrees with Faranda et al. (2016), who linked blocking events to unstable fixed points of the atmospheric dynamics, and Lucarini and Gritsun (2020), who showed in an idealized context that blocking displays anomalously high instability relative to zonal flow.

More generally, we conclude that most Atlantic-European weather regimes display relatively stable flow characteristics (low d and θ) at their maximum stage, yet relatively unstable flow (high d and θ) at onset and decay. Large decreases in d and θ systematically occur toward the maximum stage of each weather regime, even for those that display above average d and θ , whereas large increases in these metrics arise toward their decay stage. We interpret these differences as pointing to most weather regimes being states which have a physical footprint, reflected in the dynamical properties of the atmospheric flow.

Cluster analysis is obviously a valuable tool for simplifying the investigation of large amounts of data, with applications that include revealing atmospheric teleconnections (Kucharski et al., 2010), model





Figure 5. Cumulative Distribution Functions (CDFs) of the deseasonalized dynamical systems metrics (d and θ) at the *maximum* stage of the Atlantic-European weather regimes. The weather regimes considered are: Atlantic Trough (AT), Zonal Regime (ZO), Scandinavian Trough (SCTr), Atlantic Ridge (AR), European Blocking (EuBL), Scandinavian Blocking (ScBL), and Greenland Blocking (GL). The differences significant at the 5% level are shown in Table S2 in Supporting Information S1.

evaluation (Maloney et al., 2019), quantifying the impact of external forcing on climate variability (Peterson et al., 2012), and many more. The results presented here validate the relevance of weather regime clusters for such applications. For example, the dynamical systems metrics combined with cluster analysis may provide a quantitative framework for evaluating the dynamics of Atlantic-European weather regimes in climate models, thus providing the basis for studying the changes in the dynamics of weather regimes under future scenarios. Furthermore, as the dynamical systems metrics provide quantitative information on the intrinsic predictability of weather regimes, they may be used to identify windows of opportunity for subseasonal weather forecasting (Mariotti et al., 2020).

In summary, our findings support the potential of using a dynamical system view of weather regimes for improving our understanding of both weather predictions and climate projections. In particular, we argue that great potential for improvement in subseasonal weather forecasting lies in further understanding of weather regimes and their physical grounding.

Conflict of Interest

The authors declare no conflicts of interest relevant to this study.

Data Availability Statement

The analysis in this study is based on the European Centre for Medium-Range Weather Forecast (ECMWF) ERA-Interim reanalysis (https://www.ecmwf.int/en/forecasts/datasets/reanalysis-datasets/era-interim; Dee et al., 2011). The code we have used for our dynamical systems analysis is freely available at this location: https://ch.mathworks.com/matlabcentral/fileexchange/95768-attractor-local-dimension-and-local-persistence-computation. The weather regime life cycle classification is available through Grams et al. (2017).

References

Cehelsky, P., & Tung, K. K. (1987). Theories of multiple equilibria and weather regimes: A critical re-examination. Part II: Baroclinic two-layer models. *Journal of the Atmospheric Sciences*, 44(21), 3282–3303. https://doi.org/10.1175/1520-0469(1987)044<3282:tome aw>2.0.co;2

Charney, J. G., & DeVore, J. G. (1979). Multiple flow equilibria in the atmosphere and blocking. *Journal of the Atmospheric Sciences*, *36*, 1205–1216. https://doi.org/10.1175/1520-0469(1979)036<1205:MFEITA>2.0.CO;2

Dee, D. P., Uppala, S. M., Simmons, A. J., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society*, 137, 553–597. https://doi.org/10.1002/qj.828

Acknowledgments

This work was a contribution to the hydrological cycle in the Mediterranean Experiment (HyMex) and COST MEDCYCLONES: European Network for Mediterranean Cyclones in weather and climate. The ECMWF is acknowledged for granting access to the ERA-Interim data. The data analysis and visualization were partly performed using the NCAR Command Language (UCAR/NCAR/CISL/VETS, 2012). AH was funded by the German Helmholtz Association ("Changing Earth" program). GM was supported by the Swedish Research Council Vetenskapsrådet (Grant No. 2016-03724) and by the European Research Council (ERC) under the European Union's Horizon 2020 research and innovation programme (project CENÆ: compound Climate Extremes in North America and Europe: from dynamics to predictability, Grant Agreement No. 948309). The contribution of JQ and CMG was funded by the Helmholtz Association as part of the Young Investigator Group "Subseasonal Predictability: Understanding the Role of Diabatic Outflow" (SPREADOUT, Grant VH-NG-1243). JGP thanks the AXA Research Fund (https://www.axa-research.org/en/ project/joaquim-pinto) for support. This research was partially embedded in the subprojects A8 and C8 of the Transregional Collaborative Research Centre SFB/TRR 165 "Waves to Weather" (https://www.wavestoweather.de) funded by the German Research Foundation (DFG). Open access funding enabled and organized by Projekt DEAL.

- Dorrington, J., & Strommen, K. J. (2020). Jet speed variability obscures Euro-Atlantic regime structure. *Geophysical Research Letters*, 47, e2020GL087907. https://doi.org/10.1029/2020GL087907
- Falkena, S. K., de Wiljes, J., Weisheimer, A., & Shepherd, T. G. (2020). Revisiting the identification of wintertime atmospheric circulation regimes in the Euro-Atlantic sector. Quarterly Journal of the Royal Meteorological Society, 146, 2801–2814. https://doi.org/10.1002/ qj.3818
- Faranda, D., Alvarez-Castro, M. C., Messori, G., Rodrigues, D., & Yiou, P. (2019). The Hammam effect or how a warm ocean enhances large scale atmospheric predictability. *Nature Communications*, 10(1), 1316. https://doi.org/10.1038/s41467-019-09305-8
- Faranda, D., Masato, G., Moloney, N., Sato, Y., Daviaud, F., Dubrulle, B., & Yiou, P. (2016). The switching between zonal and blocked mid-latitude atmospheric circulation: A dynamical system perspective. *Climate Dynamics*, 47, 1587–1599. https://doi.org/10.1007/ s00382-015-2921-6
- Faranda, D., Messori, G., & Vannistem, S. (2019). Attractor dimension of time-averaged climate observables: Insights from a low-order ocean-atmosphere model. *Tellus A*, 71, 1554413. https://doi.org/10.1080/16000870.2018.1554413
- Faranda, D., Messori, G., & Yiou, P. (2017). Dynamical proxies of North Atlantic predictability and extremes. *Scientific Reports*, 7, 41278. https://doi.org/10.1038/srep41278
- Fereday, D. (2017). How persistent are North Atlantic-European sector weather regimes? Journal of Climate, 30(7), 2381–2394. https://doi. org/10.1175/jcli-d-16-0328.1
- Freitas, A. C. M., Freitas, J. M., & Todd, M. (2010). Hitting time statistics and extreme value theory. Probability Theory and Related Fields, 147, 675–710. https://doi.org/10.1007/s00440-009-0221-y
- Ghil, M., & Robertson, W. R. (2002). "Waves" vs. "particles" in the atmosphere's phase space: A pathway to long-range forecasting? Proceedings of the National Academy of Sciences of the United States of America, 99, 2493–2500. https://doi.org/10.1073/pnas.012580899
- Grams, C. M., Beerli, R., Pfenninger, S., Staffell, I., & Wernli, H. (2017). Balancing Europe's wind-power output through spatial deployment informed by weather regimes. *Nature Climate Change*, 7, 557–562. https://doi.org/10.1038/nclimate3338
- Hannachi, A., Straus, D. M., Franzke, C. L. E., Corti, S., & Woolings, T. (2017). Low-frequency nonlinearity and regime behaviour in the Northern Hemisphere extratropical atmosphere. *Review of Geophysics*, 55, 199–234. https://doi.org/10.1002/2015RG000509
- Hochman, A., Alpert, P., Harpaz, T., Saaroni, H., & Messori, G. (2019). A new dynamical systems perspective on atmospheric predictability: Eastern Mediterranean weather regimes as a case study. *Science Advances*, 5(6), eaau0936. https://doi.org/10.1126/sciadv.aau0936
- Hochman, A., Alpert, P., Kunin, P., Rostkier-Edelstein, D., Harpaz, T., Saaroni, H., & Messori, G. (2020). The dynamics of cyclones in the 21st century: The eastern Mediterranean as an example. *Climate Dynamics*, 54(1–2), 561–574. https://doi.org/10.1007/s00382-019-05017-3
- Hochman, A., Scher, S., Quinting, J., Pinto, J. G., & Messori, G. (2020). Dynamics and predictability of cold spells over the eastern Mediterranean. *Climate Dynamics*. https://doi.org/10.1007/s00382-020-05465-2
- Hochman, A., Scher, S., Quinting, J., Pinto, J. G., & Messori, G. (2021). A new view of heatwave dynamics and predictability over the eastern Mediterranean. *Earth System Dynamics*, *12*, 133–149. https://doi.org/10.5194/esd-12-133-2021
- Kucharski, F., Kang, I., Straus, D., & King, M. P. (2010). Teleconnections in the atmosphere and oceans. Bulletin of the American Meteorological Society, 91, 381–383. https://doi.org/10.1175/2009BAMS2834.1
- Legras, B., & Ghil, M. (1985). Persistent anomalies, blocking and variations in atmospheric predictability. *Journal of the Atmospheric Sciences*, 42, 433–471. https://doi.org/10.1175/1520-0469(1985)042<0433:PABAVI>2.0.CO;2
- Levick, R. B. M. (1949). Fifty years of English weather. Weather, 4, 206-211. https://doi.org/10.1002/j.1477-8696.1949.tb05487.x
- Lucarini, V., Faranda, D., Freitas, A. C. M., Freitas, J. M., Holland, M., Kuna, T., et al. (2016). Extremes and recurrence in dynamical systems. In *Pure and applied mathematics: A Wiley series of texts monographs and tracts* (pp. 126–172). Hoboken, NJ: Wiley. https://doi.org/10.1002/9781118632321.ch6
- Lucarini, V., Faranda, D., & Wouters, J. (2012). Universal behaviour of extreme value statistics for selected observables of dynamical systems. *Journal of Statistical Physics*, 147, 63–73. https://doi.org/10.1007/s10955-012-0468-z
- Lucarini, V., & Gritsun, A. (2020). A new mathematical framework for atmospheric blocking events. *Climate Dynamics*, 54, 575–598. https://doi.org/10.1007/s00382-019-05018-2
- Madonna, E., Li, C., Grams, C. M., & Woollings, T. (2017). The link between eddy-driven jet variability and weather regimes in the North Atlantic-European sector. Quarterly Journal of the Royal Meteorological Society, 143, 2960–2972. https://doi.org/10.1002/qj.3155
- Majda, A. J., Franzke, C. L., Fischer, A., & Crommelin, D. T. (2006). Distinct metastable atmospheric regimes despite nearly Gaussian statistics: A paradigm model. Proceedings of the National Academy of Sciences of the United States of America, 103(22), 8309–8314. https:// doi.org/10.1073/pnas.0602641103
- Maloney, E. D., Gettelman, A., Ming, Y., Neelin, J. D., Barrie, D., Mariotti, A., et al. (2019). Process-oriented evaluation of climate and weather forecasting models. *Bulletin of the American Meteorological Society*, 100, 1665–1686. https://doi.org/10.1175/BAMS-D-18-0042.1
- Mariotti, A., Baggett, C., Barnes, E. A., Becker, E., Butler, A., Collins, D. C., et al. (2020). Windows of opportunity for skillful forecasts sub seasonal to seasonal and beyond. Bulletin of the American Meteorological Society, 101(5), E608–E625. https://doi.org/10.1175/ BAMS-D-18-0326.1
- Merryfield, W. J., Baehr, J., Batté, L., Becker, E. J., Butler, A. H., Coelho, C. A. S., et al. (2020). Current and emerging developments in sub seasonal to decadal prediction. *Bulletin of the American Meteorological Society*, 101, E869–E896. https://doi.org/10.1175/ BAMS-D-19-0037.1
- Messori, G., Caballero, R., & Faranda, D. (2017). A dynamical systems approach to studying midlatitude weather extremes. *Geophysical Research Letters*, 44, 3346–3354. https://doi.org/10.1002/2017GL072879
- Messori, G., Harnik, N., Madonna, E., Lachmy, O., & Faranda, D. (2021). A dynamical systems characterization of atmospheric jet regimes. *Earth System Dynamics*, 12, 233–251. https://doi.org/10.5194/esd-2020-8
- Michel, C., & Rivière, G. (2011). The link between Rossby wave breakings and weather regime transitions. *Journal of the Atmospheric Sciences*, 68, 1730–1748. https://doi.org/10.1175/2011JAS3635.1
- Michelangeli, P. A., Vautard, R., & Legras, B. (1995). Weather regimes: Recurrence and quasi-stationary. Journal of the Atmospheric Sciences, 52(8), 1237–1256. https://doi.org/10.1175/1520-0469(1995)052<1237:WRRAQS>2.0.CO;2
- Peterson, T. C., Stott, P. A., & Herring, S. (2012). Explaining extreme events of 2011 from a climate perspective. Bulletin of the American Meteorological Society, 93, 1041–1067. https://doi.org/10.1175/BAMS-D-12-00021.1
- Rodwell, M. J., Magnusson, L., Bauer, P., Bechtold, P., Bonavita, M., Cardinali, C., et al. (2013). Characteristics of occasional poor Medium-Range Weather Forecasts for Europe. Bulletin of the American Meteorological Society, 94(9), 1393–1405. https://doi.org/10.1175/ bams-d-12-00099.1
- Süveges, M. (2007). Likelihood estimation of the extremal index. Extremes, 10(1-2), 41-55. https://doi.org/10.1007/s10687-007-0034-2

- UCAR/NCAR/CISL/VETS. (2012). The NCAR Command Language (version 6.0.0) [Software]. Boulder, CO: UCAR/NCAR/CISL/VETS. https://doi.org/10.5065/D6WD3XH5
- Vautard, R. (1990). Multiple weather regimes over the North Atlantic: Analysis of precursors and successors. Monthly Weather Review, 118, 2056–2081. https://doi.org/10.1175/1520-0493(1990)118<2056:MWROTN>2.0.CO;2
- Wallace, J. M., & Gutzler, D. S. (1981). Teleconnections in the geopotential height field during the Northern Hemisphere winter. Monthly Weather Review, 109, 784–812. https://doi.org/10.1175/1520-0493(1981)109<0784:TITGHF>2.0.CO;2
- White, C. J., Carlsen, H., Robertson, A. W., Klein, R. J., Lazo, J. K., Kumar, A., et al. (2017). Potential applications of sub seasonal-to-seasonal (S2S) predictions. *Meteorological Applications*, 24, 315–325. https://doi.org/10.1002/met.1654
- Woollings, T., Hannachi, A., & Hoskins, B. (2010). Variability of the North Atlantic eddy-driven jet stream. Quarterly Journal of the Royal Meteorological Society, 136, 856–868. https://doi.org/10.1002/qj.625
- Yoden, S. (1983). Nonlinear interactions in a two-layer, quasi-geostrophic, low-order model with topography. Journal of the Meteorological Society of Japan, 61, 1–18. https://doi.org/10.2151/jmsj1965.61.1_1